Title: An evaluation of the discriminant and predictive validity of relative social disadvantage as screening criteria for priority access to public general dental care, in Australia.

Authors:

Kelly Jones Dr (kelly.jones@adelaide.edu.au)

Version: 2 Date: 27 February 2013

Author’s response to reviews:

1. Ethics section added to the manuscript.
2. Qualification removed from the submission system